Author's response to reviews

Title: Patient safety in primary care: a survey of general practitioners in the Netherlands

Authors:

Sander Gaal (s.gaal@iq.umcn.nl)
Wim Verstappen (w.verstappen@iq.umcn.nl)
Michel Wensing (m.wensing@iq.umcn.nl)

Version: 5 Date: 19 January 2010

Author's response to reviews: see over
Cover letter

Thank you for your response. We changed the points below according to your remarks.

Formatting changes to be made:

---

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country.

Done

References: reference 8, Submitted. Articles that have not been accepted for publication should not be cited in the reference list, unless they are publicly available as a pre-print. Instead, please refer to them in the text in the following way: (Authors: Title of manuscript, submitted).

Just yesterday we received the news this article is accepted by the J Eval Clin Pract. We changed reference 8 into an in press article according to the guidelines on your website.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, 'The Title of My Prestigious Publication: A Scientific Study.' should be changed to 'The title of my prestigious publication: a scientific study'.

Done

Text: Please ensure that all text is presented in black.

Done
Tables: please remove the visible vertical lines from your tables.

Done

Typography

Some typo’s are changed. Reference 24 is added (see table 1); unfortunately this reference was accidentally deleted in a prior version.